Author's response to reviews

Title: Treatment outcome of tuberculosis patients at Gondar University Teaching Hospital, Northwest Ethiopia. A five - year retrospective study.

Authors:

Belay Tessema (bt1488@yahoo.com)
Abebe Muche (abemuche@yahoo.com)
Assegedech Bekele (assba1add@yahoo.co.uk)
Dieter Reissig (reiigo@yahoo.de)
Frank Emmrich (Frank.Emmrich@medizin.uni-leipzig.de)
Ulrich Sack (Ulrich.Sack@medizin.uni-leipzig.de)

Version: 4 Date: 23 September 2009

Author's response to reviews: see over
Author's response to reviews

Title: Treatment outcome of tuberculosis patients at Gondar University Teaching Hospital, Northwest Ethiopia. A five-year retrospective study.

Authors:

Belay Tessema (bt1488@yahoo.com)
Abebe Muche (abemuche@yahoo.com)
Assegedech Bekele (assba1add@yahoo.co.uk)
Dieter Reissig (reiiggo@yahoo.de)
Frank Emmrich (Frank.Emmrich@medizin.uni-leipzig.de)
Ulrich Sack (Ulrich.Sack@medizin.uni-leipzig.de)

Version: 4 Date: 22 September 2009

Author's response to reviews: see over
22th September 2009

The Editorial Team

BMC Public Health

MS: 2961839982776564- ‘Treatment outcome of tuberculosis patients at Gondar University Teaching Hospital, Northwest Ethiopia. A five-year retrospective study.’

Dear Editorial Team,

The following changes were made to this revised manuscript.

1) Tables were re-layout without vertical elements.

2) Key Words were removed.

3) “Materials and Methods” was renamed as 'Methods'

Sincerely,

Belay Tessema